Author's response to reviews

Title: Nonsteroidal anti-inflammatory drug use and Alzheimer's disease risk: the MIRAGE Study

Authors:

- Agustin G Yip (agyip@bu.edu)
- Robert C Green (rcgreen@bu.edu)
- Matthew Huyck (mhuyck@bu.edu)
- L Adrienne Cupples (adrienne@bu.edu)
- Lindsay A Farrer (farrer@bu.edu)

Version: 4 Date: 24 October 2004

Author's response to reviews:

1. Authors' qualifications removed from title page.
2. Affiliations formatted as requested by editors.
3. Abstract: drug to drugs; Conclusion to Conclusions
4. Section headings: unnecessary capitalisations removed
5. Authors contributions: authors' initials used
6. Tables have remained oriented in landscape (will be misaligned if in portrait)
7. Table titles: unnecessary capitalisations removed
8. Figure title: image does not include title